Author's response to reviews

Title: A study protocol: a community pharmacy-based intervention for improving the management of sleep disorders in the community settings

Authors:

Zaswiza Mohamad Noor (zaswiza.bintimohamadnoor@uqconnect.edu.au)
Alesha Smith (asmith@pharmacy.uq.edu.au)
Simon Smith (simon.smith@qut.edu.au)
Lisa Nissen (l.nissen@uq.edu.au)

Version: 2 Date: 6 August 2012

Author's response to reviews:

School of Pharmacy,
Pharmacy Australia Centre of Excellence (PACE),
The University of Queensland,
20, Cornwall Street, Wolloongabba,
Queensland 4102,
Australia.

Date: 6 August 2012

Dr Balabanova and Editorial Board editors,
BMC Health Services Research.

Dear Dr Editors,

Herewith attached is the details of the revisions in my revised manuscript:

1. We would like to clarify that we have an ethical approval from the School of Pharmacy, University of Queensland (letter attached)

2. We would like to clarify that we do not have any other publications (in BMC and other journals) which is related to this study at the moment.

3. Changes have been made to the manuscript as below:
   - the Trial Registry number has been included in the abstract (ACTRN12612000825853)
- Power sample calculation has been revised and included in the paragraph.

4. We also clarify that we do not have any grant for this study. Funding is directly from the School of Pharmacy, University of Queensland.

Thank you for your consideration of my work.

With regards

Zaswiza Mohamad Noor.